Author’s response to reviews

Title: Measuring food provision in Western Australian long day care (LDC) services: A weighed food record method/protocol at a service level.

Authors:

Ros Sambell (r.sambell@ecu.edu.au)
Ruth Wallace (r.wallace@ecu.edu.au)
Leesa Costello (l.costello@ecu.edu.au)
Johnny Lo (j.lo@ecu.edu.au)
Amanda Devine (a.devine@ecu.edu.au)

Version: 1 Date: 28 May 2019

Author’s response to reviews:

Dear Dr Collins

Thank you for the opportunity to edit the submitted manuscript. All reviewers comments have been addressed and these minor revisions are highlighted in the updated manuscript attached. In addition, the responses to the authors comments can be found in the attached Microsoft Excel document.

I wanted to advise your editorial team that I did not receive the email (nor did my collaborating authors), dated the 2nd March 2019, advising the outcomes of the review, hence the delay in my response. I thought this was worth mentioning as it would appear there has been a system error. I was able to see the progress of the review through the BMC author submission portal.

Thank you again for the opportunity and I look forward to hearing from you.

Kind regards

Ros Sambell